CSCI 1323 (Discrete Structures), Spring 2001 Homework 20

Assigned: April 26, 2001.

Due: May 5, 2001, at the start of the exam period. Not accepted late.

Credit: 20 extra credit points. You are not required to turn in any of these problems; any you turn in will be graded, and any points earned will be added to your total homework score as extra credit.

1 Problems

1. (3 points) Do problem 2 on p. 351 of the textbook. 2. (3 points) Do problem 42 on p. 357 of the textbook. 3. (3 points) Do problem 52 on p. 358 of the textbook. 4. (3 points) Do problem 2 on p. 374 of the textbook. 5. (3 points) Do problem 17 on p. 377 of the textbook. 6. (5 points) Do problem 37 on p. 378 of the textbook.